Author's response to reviews

Title: Mild gestational diabetes in pregnancy and the adipoinsular axis in babies born to mothers in the ACHOIS randomised controlled trial

Authors:

Louise K Pirc (louise.pirc@nwahs.sa.gov.au)
Julie A Owens (julie.owens@adelaide.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)
Kristyn Willson (kristyn.willson@adelaide.edu.au)
Miles J DeBlasio (miles.deblasio@adelaide.edu.au)
Jeffrey S Robinson (jeffrey.robinson@adelaide.edu.au)

Version: 5 Date: 11 April 2007

Author's response to reviews:

We have made the following formatting changes as requested:
The manuscript title and author list have been changed to mirror that of the submission system. The submission system is correctly formatted, the manuscript is not i.e. remove unnecessary capitals from the title, and state the author names fully.

A superscript number has been used to denote affiliation.

The abstract is identical in both the manuscript and submission system.